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ABSTRACT 

This thesis documents a study of implementing advanced control schemes towards bioreactor 

control applications. Bioreactor operation is a primary operation in biotherapeutics 

manufacturing which significantly impacts both the amount and quality of the product being 

manufactured. Complexity of the control strategy increases with reactor size, which may vary 

from thousands to tens of thousands of liters in commercial manufacturing. The regulatory 

initiative of Process Analytical Technology (PAT) has highlighted the need for having robust 

monitoring tools and effective control schemes that are capable of taking real time information 

about the critical quality attributes (CQA) and the critical process parameters (CPP) and 

executing immediate response as soon as a deviation occurs. However, the limited flexibility 

that present commercial software packages offer may create a hurdle. We have developed an 

integrated program using a visual programming environment for carrying out bioreactor 

operations through which various peripheral devices are interfaced. The proposed program 

facilitates real-time access to data and allows execution of control actions to follow the desired 

trajectory. 

Apparently, regulation at the cellular level is a complex phenomenon and as a result highly 

non-linear state-time profiles are used to characterize bioprocesses. In such circumstances, 

special attention must be devoted to design and to implement advanced controllers. Baker’s 

yeast fermentation process possesses an appropriate degree of complexity so that successful 

demonstration of implementing an advanced non-linear controller for this process presents a 

challenge. Consequently, by solving a control problem for this process, in turn, becomes 

equivalent to providing a solution for a range of bioprocesses.  

A model for baker’s yeast fed-batch fermentation was selected from the literature. This model 

was further examined from a system perspective to ensure that it possessed correct structure in 
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terms of controllability and observability. The model was validated by experiments. Using this 

model, a decoupled input-output linearizing controller (DIOLC) was developed. This controller 

possesses two separate control loops – substrate controller by substrate feeding rate and 

dissolved oxygen (DO) regulated by air flow rate. The performance of DIOLC was 

benchmarked against the performance of a proportional–integral–derivative (PID) controller 

under different set of conditions. A case study also demonstrated comparing performances of 

DIOLC and PID controller for biomass maximization in the baker’s yeast process. Superiority 

of DIOLC was demonstrated over PID controller due to the ability of DIOLC to negate the 

interactions between the inputs and outputs. 

A different control strategy that did not have a strong dependence on the quality of the model 

was used to control fed-batch fermentation of Pichia pastoris for the production of recombinant 

human serum albumin (rHSA). Suitability of adaptive control was examined. Using principles 

of decoupling (as used in the formulation of DIOLC), a decoupled adaptive controller (DAC) 

was implemented for this process. Results of the performances of DAC (in spite of some 

disturbances in DO) were found to be at par compared to that with PID controller. DAC was 

better at controlling substrate than DO as measurements of DO showed disturbances in the 

transition phase (from glycerol batch to methanol induction phase). As a result, augmenting 

the strengths of both DAC and DIOLC was utilized to develop a third hybrid control scheme. 

Another important aspect that has been examined is the proteolytic degradation of recombinant 

products by host-specific proteases that hinder effective production and purification of 

heterologous proteins at high cell density fermentation from yeasts such as S. cerevisiae and P. 

pastoris. We demonstrated that use of conductivity sensor as a potential PAT tool for 

monitoring real-time measurements of medium conductivity can lead to higher product titer 

with minimal protease formation.  
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सार 

यह शोध बायोरिएक्टि नियंत्रण अिपु्रयोगों की ओि उन्ित नियंत्रण योजिाओ ंको लाग ूकिि ेके एक अध्ययि का दस्तावेजीकिण 

किता ह।ै बायोरिएक्टि ऑपिेशि बायोथेिेप्यनूटक्स निर्ााण र्ें एक प्राथनर्क ऑपिेशि ह ैजो उत्पाद की र्ात्रा औि गणुवत्ता दोिों 

पि काफी प्रभाव डालता ह।ै रिएक्टि आकाि के साथ नियंत्रण िणिीनत की जनटलता बढ़ जाती ह,ै जो हजािों से वानणनययक 

उत्पादि र्ें हजािों लीटि तक नभन्ि हो सकती ह।ै प्रोसेस एिानलनटकल टेक्िोलॉजी (पीएटी) की नवनियार्क पहल र्ें र्जबतू 

निगिािी उपकिण औि प्रभावी नियंत्रण योजिाए ंहोि ेकी ज़रूित पि प्रकाश डाला गया ह ैजो र्हत्वपणूा गणुवत्ता नवशेषताओ ं

(सीक्यएू) औि र्हत्वपणूा प्रनिया र्ापदडंों (सीपीपी) के बािे र्ें वास्तनवक सर्य की जािकािी लेि ेऔि तत्काल निष्पानदत 

किि ेर्ें सक्षर् हैं। जैसे ही नवचलि होता ह,ै उतिी ही प्रनतनिया। हालांनक, सीनर्त लचीलापि जो वतार्ाि वानणनययक सॉफ्टवेयि 

संकुल प्रदाि किता ह,ै वह बाधा पैदा कि सकता ह।ै हर्ि ेबायोरिएक्टि ऑपिेशि के नलए एक नवज़अुल प्रोग्रानर्ंग परिवेश का 

उपयोग किके एक एकीकृत प्रोग्रार् नवकनसत नकया ह ैनजसके र्ाध्यर् से नवनभन्ि परिधीय नडवाइस इटंिफेस कि िह ेहैं। प्रस्तानवत 

कायािर् डेटा पि वास्तनवक सर्य तक पह चं की सनुवधा दतेा ह ैऔि वांनित प्रक्षेपवि का पालि कििे के नलए नियंत्रण कायों 

के निष्पादि की अिरु्नत दतेा ह।ै 

जानहि ह,ै सेलुलि स्ति पि नवनियर्ि एक जनटल घटिा ह ैऔि इसके परिणार्स्वरूप अत्यनधक िॉि-लीनियि स्टेट टाइर् 

प्रोफाइल का उपयोग बायोप्रोनससों को नचनित किि ेके नलए नकया जाता ह।ै ऐसी परिनस्थनतयों र्ें, उन्ित नियंत्रकों को नडजाइि 

किि ेऔि लाग ूकििे के नलए नवशेष ध्याि सर्नपात होिा चानहए। बेकि की नकण्वि प्रनिया र्ें जनटलता की एक उपयकु्त नडग्री 

ह,ै तानक इस प्रनिया के नलए एक उन्ित गैि-िेखीय नियंत्रक को लाग ूकििे का सफल प्रदशाि एक चिुौती प्रस्तुत किता ह|ै 

ितीजति, इस प्रनिया के नलए एक नियंत्रण सर्स्या को सलुझाि ेके द्वािा, कई तिह के बायोप्रोनससों के नलए एक सर्ाधाि 

प्रदाि किि ेके बिाबि बि जाता ह।ै 

बेकि'स यीस्ट के फेिर्ेंटशि प्रोसेस के नलए एक र्ॉडल को सानहत्य से चिुा गया था। इस र्ॉडल को एक नसस्टर् परिप्रके्ष्य से 

आग ेकी जांच किि ेके नलए यह सनुिनित नकया गया था नक यह नियंत्रण योग्यता औि निगिािी के र्ार्ले र्ें सही ढांचा िखता 
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ह।ै र्ॉडल प्रयोगों द्वािा र्ान्य नकया गया था। इस र्ॉडल का इस्तेर्ाल कित ेह ए, एक नडकप्लेड इिपटु-आउटपटु िेखीयरिंग 

नियंत्रक (डायओलसी) नवकनसत नकया गया था। इस नियंत्रक र्ें दो अलग-अलग नियंत्रण लूप हैं - सब्सटे्रट फीनडंग िेट द्वािा 

सब्सटे्रट कंट्रोलि औि वाय ुप्रवाह दि द्वािा नियंनत्रत ऑक्सीजि (डीओ) भंग। डायओलसी का प्रदशाि नवनभन्ि परिनस्थनतयों के 

तहत आिपुानतक-अनभन्ि-व्यतु्पन्ि (पीआईडी) नियंत्रक के प्रदशाि के नखलाफ बेंचर्ाका  था| एक र्ार्ले का अध्ययि भी 

बेकि'स यीस्ट के फेिर्ेंटशि प्रोसेस र्ें बायोर्ास के अनधकतर्किण के नलए डायओलसी औि पीआईडी नियंत्रक के प्रदशाि की 

तुलिा किता ह।ै निनवनियों औि आउटपटु के बीच बातचीत को िकािि ेके नलए डायओलसी की क्षर्ता के कािण डायओलसी 

का श्रेष्ठता पीआईडी नियंत्रक पि प्रदनशात नकया गया था। 

एक अलग नियंत्रण िणिीनत नजसकी र्ॉडल की गणुवत्ता पि एक र्जबतू निभािता िहीं थी, को पिुः संयोजक र्ािव सीिर् 

एल्बनूर्ि (आिएचएसए) के उत्पादि के नलए नपनचया पस्टोरिस के नखलाए गए बैच नकण्वि को नियंनत्रत किि ेके नलए उपयोग 

नकया गया था। अिकूुली नियंत्रण की उपयकु्तता की जांच की गई। नडकॉनप्लंग के नसद्ांतों का उपयोग (जैसा नक डायओलसी 

के निर्ााण र्ें प्रयकु्त होता ह)ै, इस प्रनिया के नलए एक नडकप्लेड अिकूुली नियंत्रक (डीएसी) लाग ूनकया गया था। डीएसी 

(डीओ र्ें कुि गड़बड़ी के बावजदू) के प्रदशाि का परिणार् पीआईडी नियंत्रक के साथ तुलिा र्ें बिाबि पाया गया डैक की 

तुलिा र्ें डीओ की तुलिा र्ें डीएसी की तुलिा र्ें सब्सटे्रट को नियंनत्रत किि ेर्ें डीएसी बेहति था। संिर्ण चिण र्ें नग्लसिॉल 

बैच से र्ेथिॉल आयि चिण तक की गड़बड़ी नदखायी गयी। परिणार्स्वरूप, डीएसी औि डायल दोिों की शनक्तयों को बढ़ाि े

के नलए तीसिी संकि नियंत्रण योजिा नवकनसत किि ेके नलए उपयोग नकया गया था। 

एक अन्य र्हत्वपणूा पहल ूकी जांच की गई ह ैजो र्ेजबाि-नवनशि प्रोटेक्शि द्वािा पिुः संयोजक उत्पादों के प्रोटीलायनटक 

नडग्रेडेशि ह,ै जो नक एस स. सनवानसए औि नपनचया पस्टोरिस जैसे खर्ीि से उच्च कोनशका घित्व नकण्वि पि प्रोटीिों के 

प्रभावी उत्पादि औि शनुद् को िोक दतेा ह।ै हर्ि ेनदखाया ह ैनक चालकता संवेदक का उपयोग र्ध्यर् चालकता के वास्तनवक 

सर्य र्ाप की निगिािी के नलए एक संभानवत पीएटी उपकिण के रूप र्ें कर् प्रोटीज गठि के साथ उच्च उत्पाद हो सकता ह।ै 
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नकसी भी बायोप्रोसैस के नियंत्रण के नलए उत्पाद नटटाि का तीव्र र्ात्रा का एक र्हत्वपणूा इिपटु ह।ै एक एकीकृत पद्नत नवकनसत 

की गई ह ैनजसर्ें लक्ष्य प्रोटीि औि कुल प्रोटीि एकाग्रता दोिों का अिरु्ाि िर्शः आंतरिक रिबोनललनवि प्रनतदीनि औि बाहिी 

फ्लोिोसेंस के उपयोग से नकया गया ह।ै प्रस्तानवत दृनिकोण दो अलग-अलग बायोथेिेप्यनूटक्स उत्पादों, र्ािव सीिर् एल्बनूर्ि 

औि गै्रन्यलूोसाइट कॉलोिी उत्तेजक कािक के नलए र्ान्य नकया गया ह,ै िर्शः नपनचया पस्टोरिस के म्यटूी औि र्ट्स के 

उपभेदों के र्ाध्यर् से व्यक्त नकया गया। प्रस्तानवत दृनिकोण तेजी से औि पीकी नपनचया पस्टोरिस नकण्वि र्ें पीएटी कायाान्वयि 

के नलए एक र्हत्वपणूा संबल होि ेका प्रदशाि नकया गया ह।ै 

इस थीनसस का उद्दशे्य नवनभन्ि चिुौनतयों का सार्िा कििा ह ैजो नवनभन्ि हाडावेयि सेंसिों के एकीकिण, पयावेक्षी कायािर् के 

नवकास, र्ॉडल नवकास औि नवनभन्ि नियंत्रण योजिाओ ंको नडजाइि किि ेऔि कायाानन्वत किि,े पीएटी आधारित प्रनिया 

नवकास औि उत्पाद आकलि उपकिणों की जांच सनहत नवनभन्ि चिुौनतयों का सार्िा कििा ह।ै हर् आशा कित ेहैं नक इि 

प्रस्तानवत िणिीनतयों औि उपकिणों का उपयोग अत्यनधक कुशल औि उत्पादक बायोफर्ाानसनटकल प्रनियाओ ंके निर्ााण के 

नलए नकया जा सकता ह।ै 
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